



Article Nash Equilibrium and Stackelberg Approach for Traffic Flow Optimization in Road Transportation Networks—A Case Study of Warsaw

Mateusz Oszczypała 🔍, Jarosław Ziółkowski *🔍, Jerzy Małachowski 🔍 and Aleksandra Lęgas 🔍

Faculty of Mechanical Engineering, Institute of Mechanics and Computational Engineering, Military University of Technology, gen. Sylwestra Kaliskiego Street 2, 00-908 Warsaw, Poland

* Correspondence: jaroslaw.ziolkowski@wat.edu.pl

Abstract: The article discusses the issue of modelling traffic flows and the transport network. Faced with an increase in the number of vehicles in road networks, the problem of congestion and the need to optimise traffic and adapt the transport infrastructure to changing demand are growing, especially in large cities. With this in mind, the authors of this publication developed a model of the road network in the north-eastern part of the Warsaw agglomeration based on the proposed algorithm. Two methods were used to optimise the distribution of traffic flows: the Nash equilibrium and the Stackelberg approach. The Nash equilibrium assumes the aim of achieving equal average times on all roads for each origin-destination (O-D) pair. This describes the state pursued by a decentralised system guided by the individual benefits of the traffic users. On the contrary, the Stackelberg approach aims to achieve optimal travel times for the entire system. The study was carried out for three scenarios that differed in the assumed traffic demand on the road network. The basic scenario assumed the average hourly traffic demand during the morning peak hour based on traffic measurements. On the other hand, the two alternative scenarios were developed as a 10% variation in traffic volumes from the baseline scenario. On the basis of the results, it was concluded that an increase in traffic volumes for all O-D pairs could result in a decrease in traffic volumes on some links of the road network. This means that the transport network is a complex system and any change in parameters can cause significant and difficult to predict changes. Therefore, the proposed approach is useful in terms of traffic forecasting for road networks under conditions of changing traffic flow volumes. Additionally, the total travel time for the entire system differed for each scenario by a percentage difference of 0.67-1.07% between the optimal solution according to the Nash equilibrium and the Stackelberg approach.

Keywords: urban road networks; optimization; traffic congestion; traffic flow; Stackelberg approach; Nash equilibrium

1. Introduction

Transport is a complex process that aims to move people and goods from the point of origin to the destination, within the existing infrastructure. The efficiency of transport processes can be determined by the value of the goal function of the optimization problem [1,2]. In practice, the fundamental indices considered when optimizing a transport process are its costs [3,4] and travel time [5–7].

Road congestion is a widespread global phenomenon resulting from high population density and increased number of motor vehicles [8]. The effects of congested road networks are a significant burden in terms of time loss, pollution, increased industrial costs, and road maintenance costs. The management of the causes and effects of congestion will become more urgent as the population grows, as congestion increases travel times and causes significant delays. From an industrial perspective, delays reduce productivity and consequently increase operating costs [9,10].



Citation: Oszczypała, M.; Ziółkowski, J.; Małachowski, J.; Legas, A. Nash Equilibrium and Stackelberg Approach for Traffic Flow Optimization in Road Transportation Networks—A Case Study of Warsaw. *Appl. Sci.* **2023**, *13*, 3085. https://doi.org/10.3390/ app13053085

Academic Editors: Roland Jachimowski and Michał Kłodawski

Received: 23 January 2023 Revised: 23 February 2023 Accepted: 24 February 2023 Published: 27 February 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Many modern vehicles are equipped with navigation devices that can determine the shortest route from origin to destination taking into account current road conditions. To reduce congestion or even avoid congestion, it is necessary for all users of the system to coordinate the route and for users to follow suggested routes. In the future, with the proliferation of autonomous vehicles, this could become a realistic solution to road congestion problems [11]. The scientific approach to traffic problems focuses on developing complex models that determine the conditions and constraints of processes in transportation systems.

A model is a qualitative and quantitative representation of a certain reality fragment (studied system) presented using a different material basis. It should reflect selected features and system properties, which are important both in terms of the studies conducted and, above all, consistent with the modelling objective(s) [12]. Creating a model reflecting selected functional aspects of a system makes studying the system become easier and more beneficial from the economic perspective. Moreover, in the case of designed systems, this enables its detailed analysis [13].

A mathematical model is a finite set of mathematical symbols and relations and strict rules for their operation, in which the symbols and relations contained in the model have an interpretation related to specific elements of the modelled part of reality [14,15]. It describes a phenomenon studied using variables that can take numerical values. Studying such a model is possible due to the use of a mathematical apparatus, which allows accurate estimation of system behaviour in assumed and allowed scenarios [16–18].

Mathematical models can be classified according to the several different criteria. Taking into consideration time, models are divided into: static—reflecting the system and its surroundings in a given state (not taking time into account) and dynamic (so-called process models)—reflecting system operation as a function of time. Another criterion is the cause-and-effect relationship, which classifies models as deterministic and stochastic. The deterministic models reflect a system in the event of complete certainty of information regarding the analyzed system. Stochastic (probabilistic) models take into account the risk and lack of information on the structure of the characteristics of the system studied, where a mathematical description of the process is reflected by random variables. The form of relationships and relations classifies models as linear or nonlinear. The linear models describe evaluation criteria, conditions, and restrictions through linear equations (inequalities) with respect to decision variables, while the nonlinear models contain decision variable functions in the form of curves of at least the second degree.

Mathematical modelling involves describing phenomena using the language of mathematics and formal logic. It is widely applied in exact sciences (including mathematics and natural sciences) and economics. Mathematical modelling is also used in the field of transport systems when designing and optimizing transport processes and systems [19–21]. The subject matter and the objective of the research are the most important factors in selecting an appropriate tool to create a model. A transport system model should reflect the modelling objective(s) and an appropriate accuracy level adapted to it. Simple models are easy to create and analyse but may provide incomplete information about the studied system and consequently lead to incorrect conclusions. On the other hand, too complex models are very time-consuming (also cost-consuming), and excessive volumes of information can alter the reliable analysis of processes and phenomena [22,23].

The increase in traffic volume in the Warsaw transport network is permanently monitored in the traffic measurements carried out by the General Directorate of National Roads and Motorways. The results of the conducted research have led to the authors' interest in transport optimization. Caban and Drozdziel [24] conducted an analysis of traffic flows in six largest cities in Poland and showed that during the peak of the morning the highest congestion occurs in Warsaw, with an average of approximately 65%. Improving road network and reducing travel time are important components of the sustainable development of large urban agglomerations. The research carried out indicates the complex nature of the problem in both scientific and practical terms.

The aim of this article is to determine the impact of the changing traffic flow on the optimal solutions, taking into consideration the travel time. The Nash equilibrium and the Stackelberg approach have been compared in a real case study. A new algorithm was proposed, on the basis of which a model of the transport system was developed. The model includes a graph of the transport network with link characteristics. Based on traffic measurements, the volume of traffic flows was determined for the three O-D pairs. The study was carried out for three scenarios: a baseline defined on the basis of traffic measurements and alternative scenarios assuming a 10% deviation. Optimal traffic flow distributions according to Nash equilibrium and the Stackelberg approach were calculated using nonlinear programming and advanced IT software. Additionally, the proposed approaches and algorithms were used to optimize the classic Nguyen–Dupuis network, and the results obtained were compared with other approaches. The novelty of this paper is the development of the new algorithm and its implementation. The real case was conducted for the optimization of the existent transport system. Process optimization using two methods (tools) allows estimating the improvement of the efficiency of the transport system operating under current and future technical conditions. Based on the results of the real traffic case study, the Price of Anarchy (PoA) was determined. Furthermore, it has been observed that an increase in the volume of traffic flows in all O-D pairs can cause a decrease in traffic volumes on certain links in the road network. The published papers did not refer to this phenomenon, which may occur during the optimization of real traffic flows. The article is divided into seven sections. In the next section, a literature review has been presented that takes into account the most relevant studies from recent years. Section 3 presents the mathematical model of the transport system and a description of the Nash equilibrium and the Stackelberg approach. Section 4 shows the results of the case study research in the three assumed scenarios. Section 5 contains a discussion of the results obtained, while Section 6 describes the limitations in applying the proposed method. Finally, the conclusions of the study were presented and directions for further research have been indicated. All notations, abbreviations and acronyms used in the text are described in Tables A1 and A2 in Appendix A.

2. Literature Review

Transport systems with a centralized decision-making process use the Stackelberg approach to optimize traffic flow distribution. The authors of [25] analyzed intermodal water-rail-road transport using the Stackelberg approach and an iterative algorithm. The optimization of an enterprise in the rail transport industry is presented in [26], using a two-tier approach to Stackelberg's game theory based on efficiency maximization.

The Nash equilibrium is used to optimize decentralized systems [5]. The authors of [27] suggested a cellular transmission model to find an optimal balance of the traffic flow density of a highway network in a highway network.

In [28], the authors analyzed a transport system through decomposing it into local subsystems managed independently. Such subsystems often cooperate or function under conditions of mutual competition. Using both the Nash equilibrium equation and the Stackelberg approach as a numerical example, this paper proves that locally determined travel prices can be both advantageous and disadvantageous for traffic users within the entire transport system, depending on the structure of the road network and movement routes. However, in the case of parallel intercity networks [29], it was demonstrated that the application of the Stackelberg approach contributes to improving the cost efficiency of a transport process relative to the solutions obtained with the Nash equilibrium.

An important element in studies of the optimization of traffic flow distribution in the transport network is determining the relationship between the value of objective functions obtained with the Nash equilibrium and the Stackelberg approach. The quotient of the objective function value for both approaches was defined as the "Price of Anarchy" (PoA) [30–34]. The authors of [35] present research on road traffic in London, Boston, and New York. The maximum PoA values for the analyzed agglomerations were 1.24–1.28,

which translates into a percentage difference in the range of 24–28%. In turn, the authors of [36] analysed the road network in Singapore, followed by determining the PoA values on individual calendar days within the study period (April 2012). The results fell within the range of 1.0–2.4.

Optimization of a single criterion transport process based on the time or cost criterion often leads to a reduction in adverse impacts on the environment [37-39]. The publication [40], based on the example of a perishable product distribution network, demonstrated that achieving an optimal solution for the transport time criterion could lead to a reduction in fuel consumption and CO₂ emissions into the atmosphere, while the authors of [41], using an example of supply chains in France, demonstrated a strong correlation between logistics costs and CO₂ emissions using only road transport.

Djenouri et al. [42] developed a graph convolution neural network to forecast traffic flow in urban transport networks. Hui et al. [43], however, developed a long-term spatialtemporal graph convolutional fusion network model (LSTFGCN), which was validated on factual data from the Caltrans performance measurement system. In [44], a deep learning approach is presented to model traffic congestion. Using traffic data from Google Maps, a hybrid support vector machine Xception was validated. The model achieved an accuracy of 97.16%. Xueting et al. [45] developed a model based on Self-Organising Feature Mapping (SOFM) to categorise urban areas according to road congestion. They conducted their study using the example of the road network in the urban area of Guiyang city.

Vandana et al. [46] proposed an approach based on Ant Colony Optimisation and Genetic Algorithm (GA) to optimise traffic light management at a road junction. In turn, Tay and Osorio [47] developed a model based on Bayesian optimisation (BO) and Gaussian process (GP), which they used to optimise the management of the traffic signal road network in New York. A solution of the problem of stop-and-go waves on a motorway was proposed in [48]. Strnad and Marsetic performed a macroscopic simulation using numerical optimization with a differential evolution algorithm. The results showed a reduction in the queue length and the number of stops.

There are many studies on the modelling and optimization of road networks and traffic flows nowadays. A large part of them focuses on traffic forecasts using a number of tools, including game theory, advanced algorithms, or neural networks. Nash equilibrium and the Stackelberg approach are mainly used for optimizing travel time, costs, and road congestion. However, there is a lack of research on the impact of changes in traffic flow volumes on the congestion that occurs in individual elements of the transport network. Therefore, in the paper a multi-scenario analysis of changes in traffic intensity in the road network was carried out. It was observed that despite the overall increase in traffic on the network under study, there was a reduction in the number of vehicles on some links.

Table 1 contains a synthetic overview of selected publications by other authors, taking into account the case study, applied optimization approaches and adopted criteria.

Paper	Case Study	Approaches	Criteria
[25]	Intermodal transport system	Stackelberg approach and iterative algorithm	Transport time
[28]	Set of local transport systems in in a competitive and cooperative environment	Nash equilibrium and Stackelberg approach	Transport costs
[49]	Road junction	Hybrid of Nash and Stackelberg approaches with Q-learning algorithm	Traffic congestion
[29]	Intercity parallel road network	Nash equilibrium and Stackelberg approach	Transport costs
[50]	Freight forwarding problem	Nash equilibrium	Transport costs
[51]	Transport network	Nash equilibrium and stochastic equilibrium	Transport time
[52]	Transport network	Nash–Cournot equilibrium	Transport costs
[5]	Transport network	Nash equilibrium	Transport costs
[42]	Urban road networks	Graph Convolutional Neural Network	Traffic flow
[43]	Urban road networks (California)	Long-term spatial-temporal graph convolutional fusion network (LSTFGCN)	Traffic flow

Table 1. Overview of publications on transport system modelling and optimization.

Paper	Case Study	Approaches	Criteria
[44]	Selected road networks in India (data from Google Maps)	Deep Learning Technique (hybrid Xception support vector machine)	Traffic congestion
[45]	Road network in urban area of Guiyang city	Self-Organizing Feature Mapping (SOFM)	Traffic congestion
[46]	Traffic light junction	Ant Colony Optimization	Traffic flow
[40]	(Ranchi, India)	and Genetic Algorithm (GA)	management
[47]	Pood natwork in Now Vork	Bayesian optimization (BO)	Traffic flow
[47]	Road network in new lork	and Gaussian process (GP)	management
[48]	Motorway	Numerical optimization with a differential evolution algorithm	Stop-and-go waves

Table 1. Cont.

In this paper, an algorithm was proposed and implemented to create a model of the transport system to solve the optimisation problem of traffic flow distribution within the road transport network. The optimization criterion is the transport time of individual road traffic users and the movement time of an entire flow within the network, over adopted origin-destination (O-D) pairs. A road network of the north-eastern part of the Warsaw agglomeration was selected to provide a practical computational example due to the significant traffic volumes therein. Three main sources of traffic flow with a single destination were identified. This was followed by the development of a mathematical transport system model, which involved defining a transport network graph, characteristics of network linear elements and traffic flow volumes together with movement direction. Transport time minimization was adopted as the single-criterion optimization objective function. Two optimization approaches were proposed. These were the Nash equilibrium and the Stackelberg approach. The considerations assumed three scenarios for traffic flow volume, namely, a baseline scenario based on traffic measurements and two alternatives. The baseline scenario corresponds to the most probable traffic situation, while the alternative scenarios assume 10% deviations of the traffic flow volumes occurring within the assumed movement routes. The traffic volume fluctuations adopted by the authors at a level of $\pm 10\%$ correspond to actual measurements presented and analyzed within many transport systems [53–55]. Optimal traffic flow distributions according to both suggested approaches were determined using IT software.

3. Transport System Modelling and Optimization

To analyse systems and optimize traffic flow distribution, transport engineering commonly uses mathematical models of transport systems. They classically focus on one of the following four elements [56], namely:

- structure—covering linear and point elements,
- characteristics—corresponding to structural properties,
- traffic flow—depicting the demand for transport services,
- traffic organization—characterizes the distribution of traffic flow in a given transport network.

The mathematical notation of a transport system model is presented by Equation (1):

$$MST = \{G, F, P, O\},\tag{1}$$

where: *MST*—transport system model, *G*—system structure graph, *F*—set of structure graph node and (or) link functions (characteristics), *P*—traffic flow, *O*—traffic organization of the transport system.

A transport system graph is made up of a set of vertices that represent point elements of the transport infrastructure and a set of links that represent linear elements. In mathematical notation, the structure graph is presented by the relationship (2) [11]:

$$G = \{W, L\},\tag{2}$$

where: *W*—set of graph nodes *G*, *L*—set of graph links *G*, which is a Cartesian product of the nodes.

The set of all nodes in the transport network can be presented in the form of a mathematical relationship (3):

$$W = \left\{ w : w = 1, \dots, \overline{W} \right\},\tag{3}$$

where: *W* is the population of the set *W*.

The location of a demand for transport services determines the decomposition of the node set into three subsets:

- $A = \{a : a \in W\}$ —origin nodes,
- $V = \{v : v \in W\}$ —intermediate nodes,
- $B = \{b : b \in W\}$ —destination nodes.

In mathematical notation, it is represented by the following relationship (4):

$$W = A \cup V \cup B. \tag{4}$$

According to the assumption adopted when modelling transport networks, all three subsets *A*, *V* and *B* are disjoint pairwise, which is expressed as a relationship (5):

$$\begin{cases}
A \cap B = \emptyset \\
A \cap V = \emptyset. \\
B \cap V = \emptyset
\end{cases}$$
(5)

Assuming that the element l_{ij} presents a transport connection between nodes w_i and w_j , it is assigned the following conventional values: 1—when there is a connection or 0—in the absence of a connection. Therefore, a set of relationships between transport network structure graph nodes can be presented as a binary matrix in the following form:

$$L = \begin{bmatrix} 0 & \cdots & l_{1N} \\ \vdots & \ddots & \vdots \\ l_{N1} & \cdots & 0 \end{bmatrix}.$$
 (6)

Points of the transport infrastructure and linear or nonlinear characteristics are described in the model of the transport system model as node and link functions. They are both determined by properties, i.e., node and link capacity, movement distance, time, and cost of movement over given links, etc. The structure graphs of the system, together with the node and link functions form a transport network (*S*), described with Equation (7):

$$S = \{G, F_W, F_L\},\tag{7}$$

where: *G*—transport system structure graph, F_W —set of functions defined on a set of nodes W, F_L —set of functions defined on a set of links *L*.

Traffic flow is a representation of people and goods moving through points and linear elements of a transport network [57,58]. The flow appearing on the origin nodes, i.e., at the entrance to a transport system, is usually identified with the market demand for transport services, while in the case of system engineering, it means the environmental impact on the studied system. However, the flow entering the transport system through the destination nodes shows the impact of the system on its surroundings. The manner in which the traffic flow passes through a transport network is called a traffic organization. This phenomenon is subject to the following requirements imposed on traffic flow in the course of executing transport system processes:

- condition of satisfying transport demand,
- condition of non-negativity of traffic flow,
- condition of traffic flow additivity,
- condition of maintained traffic flow.

Figure 1 shows the algorithm for creating a mathematical model of a transport system (TSMA) focused on optimizing the distribution of traffic flow. Defining a transport system is the initial modelling stage. It is followed by the construction of a transport network graph that indicates the origins and destinations of the traffic flow. The next step involves determining

a set of links *L* that join individual transport nodes w_{ij} . A set of movement pairs of O-D *E* and all possible travel routes P^{ab} for each pair of O-D (*a*, *b*) are determined for such a defined structure. Depending on the adopted criterion (set of criteria), each important element of the transport network must have an appropriate characteristic assigned to it. The final stage is to determine the traffic flow volumes for a selected set *E* of pairs O-D.



Figure 1. Transport system model creation algorithm (TSMA).

The structure of a model created as a result of the algorithm implemented provides grounds for an analysis of the transport system and enables optimization of the distribution of traffic flow in the transport network according to an assumed criterion (set of criteria).

3.1. Nash Equilibrium

In game theory, a Nash equilibrium refers to a non-cooperative game with two or more participants, assuming that each participant knows the strategy of the other participants. Every static game has at least one Nash equilibrium [59].

The Nash equilibrium assumes that each transport system user independently makes a decision regarding the choice of the transport route, focusing on minimizing travel time or transport costs. This corresponds to a road traffic situation, in which a vehicle driver, having current information on the volume of traffic on the individual link of the transport network (*i*, *j*) and the predicted travel time $t^{p,ab}(x^{p,ab})$ selects the route of the route of movement *p*. Road situation can be obtained on an ongoing basis (on-line) from navigation applications [60,61] or intelligent systems [62].

Traffic flows occurring at origin node $a \in A$ are assumed to be displaced to destination node $b \in B$, as described by Equation (8):

$$x^{(a,b)} = \sum_{p \in P^{ab}} x^{p,ab}, \ \forall (a,b) \in E.$$
(8)

The sum of traffic flows entering node $j \in W$ is equal to the sum of the traffic flows leaving that node, according to the relation (9):

$$\sum_{\in W} x_{ij} = \sum_{k \in W} x_{jk}, \ \forall j \in W.$$
(9)

The aggregate flow on the link (i, j) is equal to the sum of the flows in all O-D pairs $(a, b) \in E$ on all routs $p \in P^{ab}$ containing link (i, j), according to Equation (10) [11]:

$$x_{ij} = \sum_{(a,b)\in E} \sum_{p\in P^{ab}} a_{ij}^{p,ab} \delta^{p,ab} = \sum_{(a,b)\in E} \sum_{p\in P^{ab}} x^{p,ab}, \,\forall (i,j)\in L,$$
(10)

Under the non-negativity condition of the traffic flow (11) [11]:

$$x_{ij} \ge 0, \ \forall (i,j) \in L.$$

$$\tag{11}$$

The Nash equilibrium (user equilibrium) can also be interpreted as a potential stable point in a dynamic adjustment process in which individuals (road users) adapt their behaviour to that of road users, looking for strategies that give them better results [63].

The system is in a state of equilibrium, if the average transport times for all possible travel routes in relation to individual movement O-D pairs are equal. Assuming that $t^{p,ab}$ is the average movement time of the transport unit within the O-D pair (*a*, *b*) on route *p*, the Nash equilibrium equation takes the following form [27,50,64]:

$$\min\sum_{(i,j)\in L} \int_0^{x_{ij}} t_{ij}(x) \, dx,$$
(12)

$$\forall (a,b) \in E, \ \forall p \in P_{x>0}^{ab} t^{p,ab} \left(x^{p,ab} \right) = \alpha^{ab} \wedge \alpha^{ab} \to min,$$
(13)

where: *E*—set of movement pairs O-D within a transport system, P^{ab} —set of roads for movement pairs O-D (*a*, *b*), $x^{p,ab}$ —traffic flow volume on road *p* within a pair O-D (*a*, *b*), $t^{p,ab}$ —travel time of a traffic flow unit by route *p* within a pair O-D (*a*, *b*), α^{ab} —minimum travel time within a pair O-D (*a*, *b*).

If Nash equilibrium is achieved by the transport system, then any change in the path of movement by any of the traffic participants cannot be individually beneficial to them. In this situation, each traffic participant, guided by individual profit (minimum travel time), remains at his own choice.

3.2. Stackelberg Approach

In the case of transport systems where traffic flow distribution is determined by a central decision-maker (routed traffic), the total transport time may be treated as a decisive criterion for traffic flow distribution within a transport network. In the Stackelberg game, there are restrictions and conditions on the movement of traffic flows determined by the Equations (8)–(11). Therefore, given the above assumptions, the objective function with respect to the optimization of the transport system can be written as Equation (14) [26,65]:

$$T_c(X^*) = \min\left\{T_c(X) : X \in D^{dop}\right\},\tag{14}$$

where:

$$T_{c}(X) = \sum_{(a,b)\in E} \sum_{p\in P^{ab}} x^{p,ab} \times t^{p,ab} = \sum_{(i,j)\in L} x_{ij}t_{ij}(x_{ij}),$$
(15)

where: X—traffic flow distribution matrix, X*—optimal traffic flow distribution matrix, $T_c(X)$ —criterion function (total traffic flow movement time), D^{dop} —set of permissible solutions, $x^{p,ab}$ —traffic flow volume on road p within a O-D pair (a, b), $t^{p,ab}$ —movement time for a traffic flow unit on road p within the O-D pair (a, b).

In road traffic engineering, optimization based on the minimum total time criterion is called the Stackelberg traffic flow distribution or system optimum (SO) solution. Optimization is carried out according to the criterion of equal end times in order to achieve this state of a transport system. For each pair of transport O-D (*a*, *b*), the marginal times $m^{p,ab}$ for all movement routes $p \in P$ should be mutually equal, as described by the relationship (16) [25,28,29,66]:

$$\forall (a,b) \in E, \ \forall p,q \in P^{ab} \ m^{p,ab}(x) = m^{q,ab}(x).$$
(16)

The marginal time for the route p is calculated as the sum of the marginal times of the links on the route p, as presented in Equation (17):

$$m^{p,ab}(x) = \sum_{(i,j)\in p} m_{ij}(x_{ij}).$$
 (17)

And the marginal time for the link (i, j) is equal to the derivative of the product of the average travel time through link (i, j) and the size of the traffic flow on link (i, j), which is described by the Equation (18):

$$m_{ij}(x_{ij}) = \frac{d\left[t_{ij}(x_{ij}) \cdot x_{ij}\right]}{dx_{ij}}.$$
(18)

If the equality of marginal times in O-D pairs is satisfied, then a change of route by any vehicle would result in a loss for the whole system, moving the value of the objective function away from the optimal value.

3.3. Description of Research

Figure 2 shows a schematic of the methodology of the research conducted. After developing the transport network model, three scenarios were defined for the size of traffic flows. The all the constraints and conditions that the solutions must meet were determined. For the Nash equilibrium and the Stackelberg approach, the corresponding objective functions were established. Using non-linear programming, the optimal solutions for both methods were determined. On the basis of these, the traffic congestion on the individual links in the road network was calculated. Differences in total travel times were used to calculate the Price of Anarchy.



Figure 2. Flowchart of the research.

4. Results

This section presents the results of a study on optimizing traffic flows in relation to a real case study. A road network model of the north-eastern part of the Warsaw agglomeration was developed according to the proposed methodology. The optimal solutions were then determined for three scenarios of traffic flow volume according to Nash equilibrium and the Stackelberg approach.

4.1. Case Study

The presented optimization approaches have been applied to a solution to an actual road transport problem, occurring in the north-eastern part of the Warsaw agglomeration. It is an area that often experiences the phenomenon of traffic congestion caused by heavy traffic flows during the morning and afternoon rush hours. The lack of a well-developed road network in relation to the number of traffic users implies the need to search for new and better solutions, assuming appropriate limiting conditions. Figure 3 shows a road map with the main nodes plotted of the analyzed transport system.



Figure 3. Road map with road nodes plotted.

Figure 4 shows a graph of the analysed transport network. The transport network consists of eight nodes, three of which are the origin nodes, and one is the traffic flow destination node. Three traffic flows marked as x_1 , x_2 and x_3 are found at the network entrance. Vectors were used to determine the direction of movement of allowed traffic flows along nine transport links. Traffic flow marked as x_8 , which is the sum of flows in all origin nodes, is headed for the destination node.



Figure 4. Transport network graph.

A detailed description of the transport network can be found in Table 2. There are three movement O-D pairs within the analyzed system: (1,8), (2,8) and (3,8). The transport network graph for each O-D pair was used to determine the sets of possible routes composed of links L_{ij} .

Table 2. Characteristics of transport network.

Element of Model	Mathematical Notation
Structure	$G = \langle W, L \rangle$
Set of nodes	$W = \{1, 2, 3, 4, 5, 6, 7, 8\}$
Set of origin nodes	$A = \{1, 2, 3\}$
Set of intermediate nodes	$V = \{4, 5, 6, 7\}$
Set of destination nodes	$B = \{8\}$
Set of O-D pairs	$E = \{(1,8), (2,8), (3,8)\}$
Set of links	$L = \{(1,4), (2,5), (3,6), (4,7), (4,8), (5,4), (5,6), (6,7), (7,8)\}$
Set of routes for O-D pair $(1,8)$	$P^{18} = \{p_1, p_2\} = \{<(1,4), (4,8)>, <(1,4), (4,7), (7,8)>\}$
Set of routes for O-D pair (2,8)	$P^{28} = \{p_3, p_4, p_5\} = \{<(2,5), (5,4), (4,8)>, <(2,5), (5,4), (4,7), (7,8)>, <(2,5), (5,6), (6,7), (7,8)>\}$
Set of routes for O-D pair (3,8)	$P^{38} = \{p_6\} = \{<(3,6), (6,7), (7,8)>\}$

Each link in the transport network has been assigned its time characteristics, which determine the travel time through a given section of the road, depending on the flow of traffic. Increased traffic volume leads to an elevated average travel time through a transport link and a significant reduction in the average travel speed (especially in bottlenecks), which in consequence may result in congestion. Equation (19) was used to present the characteristic time matrix of the analyzed road network.

For the traffic flow x_{ij} expressed in thousands of vehicles, the characteristics of individual links in the transport network are shown in the form of a system of Equations (20):

$$\begin{cases} t_{14}(x_{14}) = 5 + 0.05x_{14} + 0.025x_{14}^2 \\ t_{25}(x_{25}) = 10 + 0.04x_{25} + 0.065x_{25}^2 \\ t_{36}(x_{36}) = 6 + 0.03x_{36} + 0.025x_{36}^2 \\ t_{47}(x_{47}) = 4 + 0.04x_{47} + 0.04x_{47}^2 \\ t_{48}(x_{48}) = 5 + 0.08x_{48} + 0.125x_{48}^2 \\ t_{54}(x_{54}) = 4 + 0.035x_{54} + 0.02x_{54}^2 \\ t_{56}(x_{56}) = 5 + 0.035x_{56} + 0.04x_{56}^2 \\ t_{67}(x_{67}) = 1 + 0.035x_{67} + 0.03x_{67}^2 \\ t_{78}(x_{78}) = 2 + 0.025x_{78} + 0.035x_{78}^2 \end{cases}$$
(20)

The set of permissible solutions is determined by the conditions to be satisfied by the distribution of the traffic flow within a transport network, which were presented through systems of equations and inequalities (21)–(24):

• condition of satisfying transport demand (21), assuming that the total traffic flow within a given O-D pair moves entirely along identified transport network roads:

$$\begin{cases} x_1 = x_{1,18} + x_{2,18} \\ x_2 = x_{3,28} + x_{4,28} + x_{5,28}, \\ x_3 = x_{6,38} \end{cases}$$
(21)

• condition of traffic flow non-negativity (22):

$$x_{14} \ge 0
x_{25} \ge 0
x_{36} \ge 0
x_{47} \ge 0
x_{48} \ge 0,$$

$$x_{54} \ge 0
x_{56} \ge 0
x_{67} \ge 0
x_{72} \ge 0$$
(22)

• condition of traffic flow additivity (23), according to which traffic flows along various O-D pairs and routes add together on common transport links:

$$x_{14} = x_{1,18} + x_{2,18}$$

$$x_{25} = x_{3,28} + x_{4,28} + x_{5,28}$$

$$x_{36} = x_{6,38}$$

$$x_{47} = x_{2,18} + x_{4,28}$$

$$x_{48} = x_{1,18} + x_{3,28}$$

$$x_{54} = x_{3,28} + x_{4,28}$$

$$x_{56} = x_{5,28}$$

$$x_{67} = x_{5,28} + x_{6,38}$$

$$x_{78} = x_{2,18} + x_{4,28} + x_{5,28} + x_{6,38}$$
(23)

• condition of maintained traffic flow (24), which assumes that the sum of all flows entering a node is equal to the sum of flows exiting this node:

$$\begin{cases}
x_1 = x_{14} \\
x_2 = x_{25} \\
x_3 = x_{36} \\
x_{14} + x_{54} = x_{47} + x_{48} \\
x_{25} = x_{54} + x_{56} \\
x_{36} + x_{56} = x_{67} \\
x_{47} + x_{67} = x_{78} \\
x_{48} + x_{78} = x_8
\end{cases}$$
(24)

4.2. Optimization with the Nash and Stackelberg Approaches

The Nash equilibrium is based on the assumption of equal average movement times per vehicle during transport operations performed within a transport network. The average travel time through route p is equal to the total of average travel times through the links that make up this link. The average travel times within the analyzed transport network are expressed by the relationship (25):

$$t_{1,18}(x) = t_{14}(x_{14}) + t_{48}(x_{48})$$

$$t_{2,18}(x) = t_{14}(x_{14}) + t_{47}(x_{47}) + t_{78}(x_{78})$$

$$t_{3,28}(x) = t_{25}(x_{25}) + t_{54}(x_{54}) + t_{48}(x_{48})$$

$$t_{4,28}(x) = t_{25}(x_{25}) + t_{54}(x_{54}) + t_{47}(x_{47}) + t_{78}(x_{78})$$

$$t_{5,28}(x) = t_{25}(x_{25}) + t_{56}(x_{56}) + t_{67}(x_{67}) + t_{78}(x_{78})$$

$$t_{6,38}(x) = t_{36}(x_{36}) + t_{67}(x_{67}) + t_{78}(x_{78})$$
(25)

For the travel O-D pair (1,8) there are two routes p_1 and p_2 , while for the O-D pair (2,8), three possible routes p_3 , p_4 and p_5 were established. The state of equilibrium is achieved if the following conditions are met (26):

$$\begin{cases} t_{1,18}(x) = t_{2,18}(x) \\ t_{3,28}(x) = t_{4,28}(x) = t_{5,28}(x) \end{cases}$$
(26)

The O-D pair (3,8) has only one possible movement route, resulting in the lack of equilibrium for flows moving between nodes 3 and 8. The equilibrium state described by the system of Equations (26) corresponds to a situation in which a change in the route of movement of any participant in the traffic would result in an extended travel time.

To perform the optimization using the Stackelberg approach, the marginal times $m_{ij}(x_{ij})$ for individual links of the road network were calculated. They are presented in the form of relationship (27):

$$\begin{cases} m_{14}(x_{14}) = \frac{d[t_{14}(x_{14}) \cdot x_{14}]}{dx_{14}} = 5 + 0.1x_{14} + 0.075x_{14}^2 \\ m_{25}(x_{25}) = \frac{d[t_{25}(x_{25}) \cdot x_{25}]}{dx_{25}} = 10 + 0.08x_{25} + 0.195x_{25}^2 \\ m_{36}(x_{36}) = \frac{d[t_{36}(x_{36}) \cdot x_{36}]}{dx_{36}} = 6 + 0.06x_{36} + 0.075x_{36}^2 \\ m_{47}(x_{47}) = \frac{d[t_{47}(x_{47}) \cdot x_{47}]}{dx_{47}} = 4 + 0.08x_{47} + 0.12x_{47}^2 \\ m_{48}(x_{48}) = \frac{d[t_{48}(x_{48}) \cdot x_{48}]}{dx_{48}} = 5 + 0.16x_{48} + 0.375x_{48}^2 \\ m_{54}(x_{54}) = \frac{d[t_{54}(x_{54}) \cdot x_{54}]}{dx_{56}} = 4 + 0.07x_{54} + 0.06x_{54}^2 \\ m_{56}(x_{56}) = \frac{d[t_{56}(x_{56}) \cdot x_{56}]}{dx_{56}} = 5 + 0.07x_{56} + 0.12x_{56}^2 \\ m_{67}(x_{67}) = \frac{d[t_{67}(x_{67}) \cdot x_{67}]}{dx_{67}} = 1 + 0.07x_{67} + 0.09x_{67}^2 \\ m_{78}(x_{78}) = \frac{d[t_{78}(x_{78}) \cdot x_{78}]}{dx_{78}} = 2 + 0.05x_{78} + 0.105x_{78}^2 \end{cases}$$

The Stackelberg approach assumes the presence of a minimum total movement time with equal marginal travel times on individual routes in the transport network routes. The marginal travel time through route p is equal to the total marginal travel times through the links comprising route p. Marginal travel times within the analyzed transport network are expressed by the relationship (28):

$$\begin{cases} m_{1,18}(x) = m_{14}(x_{14}) + m_{48}(x_{48}) \\ m_{2,18}(x) = m_{14}(x_{14}) + m_{47}(x_{47}) + m_{78}(x_{78}) \\ m_{3,28}(x) = m_{25}(x_{25}) + m_{54}(x_{54}) + m_{48}(x_{48}) \\ m_{4,28}(x) = m_{25}(x_{25}) + m_{54}(x_{54}) + m_{47}(x_{47}) + m_{78}(x_{78}) \\ m_{5,28}(x) = m_{25}(x_{25}) + m_{56}(x_{56}) + m_{67}(x_{67}) + m_{78}(x_{78}) \\ m_{6,38}(x) = m_{36}(x_{36}) + m_{67}(x_{67}) + m_{78}(x_{78}) \end{cases}$$
(28)

The condition of equal marginal times for the analyzed road network is presented by the system of Equations (29):

$$m_{1,18}(x) = m_{2,18}(x) m_{3,28}(x) = m_{4,28}(x) = m_{5,28}(x)$$
(29)

4.3. Baseline Scenario #1

The first to be analyzed was the baseline scenario, in which the traffic flow volumes that occur for individual O-D pairs during morning rush hours were estimated based on the results of the General Traffic Measurement (GPR) implemented by the General Directorate of National Roads and Motorways (GDDKiA) in the years 2020–2021. For movement O-D pairs (1,8), (2,8) and (3,8), the transport demand values amount to 4000, 5000 and 7000 veh./h, respectively.

Calculations for both optimization approaches were performed with computer assistance using LINGO 18.0 software (Lindo Systems, Chicago, USA). Table 3 lists the results obtained for the individual routes of the transport network. According to the assumptions of the Nash equilibrium, the travel times for each available route p within each O-D pair (a, b) are equal. A transport system reaches an equilibrium state with the traffic flow distribution shown in the column "Nash" in Table 3. A movement O-D pair (3,8) may be implemented only through one route p_6 , therefore the entire traffic flow must be moved through route p_6 , regardless of the optimization approach.

Table 3. Results of optimization with the Nash and Stackelberg approaches for movement routes in scenario #1.

O-D Pair	Route	Traffic Flow Volume ($ imes 10^3$ veh.)		Average Travel Time (Min)		Marginal Travel Time (Min)
		Nash	Stackelberg	Nash	Stackelberg	Stackelberg
(1.8)	p_1	4.000	4.000	15.54	15.32	28.30
(1,0)	p_2	0.000	0.000	-	-	-
	p_3	1.973	1.835	26.14	26.19	39.73
(2,8)	p_4	1.610	3.026	26.14	26.83	39.73
	p_5	1.417	0.139	26.14	25.48	39.73
(3,8)	p_6	7.000	7.000	16.62	16.09	29.54

By translating the volume of traffic flow within individual routes of the O-D pairs in question, the authors calculated the volumes of traffic within the links of the road network. The results are listed in Table 4 and presented graphically in Figures 5 and 6.

Link	Traffic Flow Vo	olume (×10 ³ veh.)	Average Travel Time (Min)		
Link	Nash	Stackelberg	Nash	Stackelberg	
(1,4)	4.000	4.000	5.60	5.60	
(2,5)	5.000	5.000	11.83	11.83	
(3,6)	7.000	7.000	7.44	7.44	
(4,7)	1.610	3.026	4.17	4.49	
(4,8)	5.973	5.835	9.94	9.72	
(5,4)	3.583	4.861	4.38	4.64	
(5,6)	1.417	0.139	5.13	5.01	
(6,7)	8.417	7.139	3.42	2.78	
(7,8)	10.027	10.165	5.77	5.87	

Table 4. Results of optimization with the Nash and Stackelberg approaches for road network links in scenario #1.

Traffic flow volume and travel time through links (1,4), (2,5) and (3,6) have the same values in the case of the Nash equilibrium and the Stackelberg approach. This results from the transport network structure, which implies the need to send all the traffic flow originating in nodes 1, 2 and 3, respectively, through these three links. The linear element most loaded of the road network is link (7,8), for which the traffic flow within the analyzed system exceeds 10,000 veh./h. On the other hand, the lowest volume is experienced in link (5,6). In the case of the Nash equilibrium, it is more than ten times higher than in the case of the Stackelberg approach traffic distribution.

Road congestion values were calculated based on travel time through individual road network links obtained for individual solutions under the assumptions of the Nash and Stackelberg approaches. The level of congestion was calculated as a quotient of the travel time through the link (i,j) under specific traffic conditions and the travel time through this link under minimum traffic conditions. The authors calculated the an increase in travel time relative to the conditions of fully free traffic. The least favorable road conditions were observed for link (6,7), in the case of Nash equilibrium, where the transport time was more than 240% longer than under standard conditions. Regardless of the optimization approach,

12 10 $(\times 10^3 \text{ vehicles})$ 8 6 4 2 0 x25 x56 x36 x48 x54 x67 x78 x14 x47 Nash Stackelberg

the lowest level of traffic congestion occurred for link (5,6), where the travel time increased by less than 3%. The results for all links in the road network are summarized in Table 5.

Figure 5. Comparison of traffic volume on individual links to road network for scenario #1.



Figure 6. Compared travel times through individual links to road network for scenario #1.

Link	Congestion		Travel Time Increase (%)	
Link	Nash	Stackelberg	Nash	Stackelberg
(1,4)	1.12	1.12	12.00	12.00
(2,5)	1.18	1.18	18.25	18.25
(3,6)	1.24	1.24	23.92	23.92
(4,7)	1.04	1.12	4.20	12.18
(4,8)	1.99	1.94	98.75	94.44
(5,4)	1.10	1.16	9.55	16.07
(5,6)	1.03	1.00	2.60	0.11
(6,7)	3.42	2.78	242.02	177.89
(7,8)	2.88	2.94	188.48	193.54

Table 5. Traffic congestion in transport network links for scenario #1.

4.4. Alternative Scenario #2

The alternative scenario #2 is a hypothetical road situation, which assumes a decrease in traffic flow volumes for all movement O-D pairs by 10%, relative to the baseline scenario. In this case, the transport demand takes the following respective values: 3600, 4500 and 6300 veh./h. The results of optimization with the Nash and Stackelberg approaches are summarized in Tables 6 and 7 and illustrated in Figures 7 and 8.

O-D Pair	Route	Traffic Flow Volume ($\times 10^3$ veh.)		Average Travel Time (Min)		Marginal Travel Time (Min)
		Nash	Stackelberg	Nash	Stackelberg	Stackelberg
(1.9)	p_1	2.585	2.376	14.63	14.39	25.56
(1,8)	p_2	1.015	1.224	14.63	15.02	25.56
	p_3	2.846	2.886	24.88	24.88	35.90
(2,8)	p_4	0.000	1.346	-	25.52	35.90
	p_5	1.654	0.268	24.88	24.18	35.90
(3,8)	<i>p</i> ₆	6.300	6.300	15.40	14.86	25.92

Table 6. Results of optimization with the Nash and Stackelberg approaches for movement routes in scenario #2.

Table 7. Results of optimization with the Nash and Stackelberg approaches for road network links in scenario #2.

	Traffic Flow Ve	olume (×10 ³ veh.)	Average Travel Time (Min)	
Link	Nash	Stackelberg	Nash	Stackelberg
(1,4)	3.600	3.600	5.50	5.50
(2,5)	4.500	4.500	11.50	11.50
(3,6)	6.300	6.300	7.18	7.18
(4,7)	1.015	2.570	4.08	4.37
(4,8)	5.431	5.262	9.12	8.88
(5,4)	2.846	4.232	4.26	4.51
(5,6)	1.654	0.268	5.17	5.01
(6,7)	7.954	6.568	3.18	2.52
(7,8)	8.969	9.138	5.04	5.15



Figure 7. Traffic volume compared in individual links to road network for scenario #2. (Nash and Stackelberg differences (%) represent the percentage increase in traffic volumes on each link compared to the results of scenario #1).



Figure 8. Compared travel times through individual links to road network for scenario #2. (Nash and Stackelberg differences (%) represent the percentage increase in travel time through each link compared to the results of scenario #1).

The number of vehicles traveling on the links to the starting nodes is strictly determined by the value of demand for transportation on each route. It should be stressed that, in the case of the analyzed system, they do not depend on the applied traffic flow distribution approach. Significant differences in traffic flow depended on the applied optimization approach occurring in links (4,7), (5,4), (5,6), and (6,7). Regardless of the applied approach (Nash, Stackelberg), the most loaded road section in both cases was on link (7,8). On the other hand, the least loaded is link (4,7) for the solution obtained with the Nash approach and link (5,6) in the case of the Stackelberg approach.

Despite the reduced transport demand for all O-D pairs by 10% compared to scenario #1, the travel time through link (5,6) for the solution using the Nash approach increased slightly, from a level of 5.13 (min) to 5.17 (min), which translates to a relative difference of 0.73%. This is caused by an increase in link traffic volume by 16.68%. In turn, when it comes to the results obtained with the Stackelberg approach, a vehicle number increased within the link (5,6) by more than 92% resulting in the travel time being extended by only 0.13%. The increase in the volume of traffic within transport links, for relatively low levels of the absolute number of vehicles, leads to minor differences in the average travel times for these links.

Table 8 shows the impact of the traffic flow distribution on the state of traffic congestion for scenario #2. The highest congestion rate was obtained for links (6,7) and (7,8), while in the case of the remaining segment of the road network this value did not exceed 2 for the results obtained regardless of the optimization approach adopted. It should be emphasized the Stackelberg approach allows a better balance of the value of the congestion rate by reducing the difference between its maximum and minimum values within the road network.

	Con	gestion	Travel Time Increase (%)	
Link	Nash	Stackelberg	Nash	Stackelberg
(1,4)	1.10	1.10	10.08	10.08
(2,5)	1.15	1.15	14.96	14.96
(3,6)	1.20	1.20	19.69	19.69
(4,7)	1.02	1.09	2.05	9.18
(4,8)	1.82	1.78	82.43	77.63
(5,4)	1.07	1.13	6.54	12.66
(5,6)	1.03	1.00	3.35	0.24
(6,7)	3.18	2.52	217.62	152.40
(7,8)	2.52	2.58	151.98	157.56

Table 8. Traffic congestion on transport network links for scenario #2.

4.5. Alternative Scenario #3

The third scenario assumes an increase in transport demand for each analyzed O-D pair by 10%, relative to the baseline scenario. This corresponds to traffic flow values of: 4400, 5500 and 7700 veh./h. The optimization results with the Nash and Stackelberg approaches are summarized in Tables 9 and 10, and illustrated in Figures 9 and 10.

Table 9. Results of optimization with the Nash and Stackelberg approaches for movement routes in scenario #3.

O-D Pair	Route	Traffic Flow Volume ($ imes 10^3$ veh.)		Average Travel Time (Min)		Marginal Travel Time (Min)
		Nash	Stackelberg	Nash	Stackelberg	Stackelberg
(1.8)	p_1	4.400	4.400	16.55	16.35	31.32
(1,8)	p_2	0.000	0.000	-	-	-
	p_3	2.125	2.009	27.55	27.63	43.95
(2,8)	p_4	2.167	3.470	27.55	28.26	43.95
	p_5	1.208	0.021	27.55	26.91	43.95
(3,8)	<i>p</i> ₆	7.700	7.700	17.98	17.43	33.52

Table 10. Results of optimization with the Nash and Stackelberg approaches for road network links in scenario #3.

T * . 1	Traffic Flow Vo	olume (×10 ³ veh.)	Average Travel Time (Min)	
Link	Nash	Stackelberg	Nash	Stackelberg
(1,4)	4.400	4.400	5.70	5.70
(2,5)	5.500	5.500	12.19	12.19
(3,6)	7.700	7.700	7.71	7.71
(4,7)	2.167	3.470	4.27	4.62
(4,8)	6.525	6.409	10.84	10.65
(5,4)	4.292	5.479	4.52	4.79
(5,6)	1.208	0.021	5.10	5.00
(6,7)	8.908	7.721	3.69	3.06
(7,8)	11.075	11.191	6.57	6.66



Figure 9. Traffic volume compared on individual links to the road network for scenario #3. (Nash and Stackelberg differences (%) represent the percentage increase in traffic volumes on each link compared to the results of scenario #1).



Figure 10. Compared travel times through individual links to the road network for scenario #3. (Nash and Stackelberg differences (%) represent the percentage increase in travel time through each link compared to the results of scenario #1).

In the case of the analyzed system, an increased transport demand for all O-D pairs of movement entails a decreased link load (5,6). This is caused by the increased sensitivity of the travel time through link (6,7) together with the increased volume of traffic. The higher volume of traffic flow results in route p_5 of the O-D pair (2,8) becoming less competitive in terms of travel time relative to routes p_3 and p_4 . This is caused by the traffic flow that moves along the O-D pair (3,8), for which there is also a route within the analyzed transport system.

Table 11 shows the results of traffic congestion analysis for scenario #3. Similar to scenarios #1 and #2, the highest congestion rate values were achieved by links (6,7) and (7,8). The average travel times for these sections of the road are greater by more than 200% in a free traffic situation. Congestion with a rate beyond 2 appeared in this scenario also on link (4,8). In turn, despite a growth in transport demand, the situation of the road on other links still did not deteriorate considerably.

T • 1	Con	gestion	Travel Time Increase (%)		
Link	Nash	Stackelberg	Nash	Stackelberg	
(1,4)	1.14	1.14	14.08	14.08	
(2,5)	1.22	1.22	21.86	21.86	
(3,6)	1.29	1.29	28.55	28.55	
(4,7)	1.07	1.16	6.86	15.51	
(4,8)	2.17	2.13	116.88	112.96	
(5,4)	1.13	1.20	12.97	19.80	
(5,6)	1.02	1.00	2.01	0.01	
(6,7)	3.69	3.06	269.24	205.86	
(7,8)	3.28	3.33	228.49	233.14	

Table 11. Traffic congestion in links of transport network for scenario #3.

4.6. Compared Optimization Results for Three Scenarios

The last stage of the research was to determine the value of the criterion for the entire transport system. The total travel time T_c for units of all traffic flow within a road network is expressed with the Equation (30):

$$T_c = \sum_{(i,j)\in L} x_{ij} t_{ij}(x_{ij}).$$
(30)

The results of the calculations for the three scenarios of traffic flow volume analyzed in the transport system are shown as a graph in Figure 11. The Stackelberg approach allowed

to achieve more favourable solutions for the entire transport system by 0.67–1.07%, while the difference for the baseline scenario #1 was 0.84%, which means that the PoA value was 1.0084. Furthermore, in the case of scenario #2, where traffic volumes were less than 10% relative to the baseline scenario #1, the difference in total times T_c between the solutions obtained using the Stackelberg approach and the Nash equilibrium was the highest.



Figure 11. Compared total travel times of the entire traffic flow for the three scenarios. (The difference (%) represents the percentage increase in total travel time between the results of the Stackelberg approach and the Nash equilibrium for each scenario).

5. Discussion

For the case study analyzed for three scenarios, considerations were made for the traffic flow distribution according to the Nash equilibrium and the Stackelberg approach. Computational analyses resulted in the determination of the optimum size of traffic flows on the links of the transport network. The criteria of equal average times and equal marginal times were met for all O-D pairs in the three scenarios.

An increase in traffic flows for each O-D pair in the transport system may result in a reduction in traffic flows on certain links in the road network. For the case study presented, this situation occurred at link (5,6). With an overall increase in traffic flows, it becomes less attractive to travel for vehicles moving from origin node 2 to destination node 8. The selection of node (5,6) for travel in the O-D pair (2,8) results in the need to travel also through node (6,7), which is loaded with a large traffic flow moving in the O-D pair (3,8). Therefore, the p_5 route is more attractive when traffic volumes are low in the transportation network.

Analysis of road congestion as an increase in travel time through transport links identifies the weakest elements of the road network, which have the greatest impact on the decrease in efficiency of transport processes. In the model for the northern part of Warsaw's road network, the greatest road congestion occurs on links (4,8), (6,7) and (7,8). According to the Nash equilibrium for scenario #3, there was a 269.24% increase in travel time on link (6,7). For the baseline scenario, the average congestion levels for the analyzed network were 66.64% and 60.93% for the Nash equilibrium and the Stackelberg approach, respectively. Relating these results to studies by other authors [24], it was found that the congestion for the user equilibrium condition obtained using the developed model corresponds to the level determined by traffic measurements.

Despite focusing on individual benefits for road users, the Nash equilibrium approach turned out to be just slightly less effective compared to minimizing the travel time of the entire traffic flow. The total transport time in the case of the Stackelberg approach was more favourable by approximately 1% compared to the Nash equilibrium approach. It should also be stressed that the time saved can contribute to potential benefits for the environment, due to lower energy consumption and reduced CO_2 emissions to the atmosphere. The

aim of optimizing traffic flow distribution should be to minimize travel time and energy consumption. Regarding the findings of other authors [35,36], it should be highlighted that the PoA for the Warsaw case study is at a significantly lower value.

Table 12 summarises the results obtained for the Nash and Stackelberg approaches and the results from the literature for the Nguyen–Dupuis network. The graph of the Nguyen–Dupuis network is shown in Appendix D in Figure 1A. Compared to the algorithm proposed by Nguyen and Dupuis [67], both the Nash equilibrium and the Stackelberg approach obtained better results. The model proposed by Xu et al. [68] achieved shorter travel times. The approaches considered in the articles [69–71] usurped significantly higher travel times for the whole network due to additional restrictions introduced by the authors. In contrast, the decision support model for infrastructure investment issues proposed by Lin [72], in the absence of any investment, was less effective in optimising travel times.

Table 12. Results of optimization on the Nash and Stackelberg approaches in the Nguyen–Dupuis network.

This Paper		Literature		
Results	Approaches	Results	Approaches	
83 228.36	Nash equilibrium	85,028.14 [67]	Nguyen-Dupuis algorithm	
82 772.50	Stackelberg approach	79,290 [68]	EC-TEP model	
	11	80,269 [68] 194,025 [69] ~240,000 [70] ~180,000 [71] ~(75,000–91,000) [72]	EC-TEP model with environmental constrains Multi-class multibehaviour equilibrium Generalized Nash equilibrium problem Bush–Mosteller reinforcement learning Bi-level model for investment decision making	

Table 13 compares the Nash equilibrium and the Stackelberg approach in terms of objective function, constraints and conditions, advantages and disadvantages. The Nash equilibrium aims at minimizing travel time for individual users. This approach reflects the current traffic situation and describes the mutual dependencies of the route choices. However, it may result in time losses for the system as a whole and an increase in the level of congestion at certain links in the network. On the contrary, minimizing the system travel time in the Stackelberg approach requires a centralized decision-making system and the compliance of all traffic users. The advantages of this method are not only a reduction in total travel time, but also a reduction in the average level of congestion in the road network.

 Table 13. Comparison of Nash equilibrium and Stackelberg approach.

Approach	Objective Function	Constrains and Conditions	Advantages	Disadvantages
Nash equilibrium	Minimization of individual users travel time	 Decentralised system Condition of satisfying transport demand Conditions of non-negativity, additivity, maintaining of traffic flow 	 Description of present situation in the road networks Reflects the mutual influence of individual traffic users on route choice 	 Causes losses for the entire system (PoA) May cause high levels of congestion on certain links in the network
Stackelberg approach	Minimization of system total travel time	 Centralised system Condition of satisfying transport demand Conditions of non-negativity, additivity, maintaining of traffic flow 	 Reduces travel time for the whole system compared to a decentralised system Reduces the average level of congestion in the road network 	 Requires a decision-making system and the compliance of all traffic users May increase travel time for some traffic users

6. Limitations

The road transport system aims to maintain a state consistent with the Nash equilibrium, provided that all road users have up-to-date and reliable information on the traffic situation and make decisions based on them. On the other hand, achieving an optimal state in line with the Stackelberg approach requires mutual cooperation of individual decisionmakers or a centralized IT system that makes decisions with respect to selecting travel routes for the entire traffic flow. In the case of the current organization of the road transport system, implementing optimization based on equal marginal times is impossible. However, in future systems equipped with fully automatic vehicles, the Stackelberg approach can be useful in solving road traffic problems.

Furthermore, the computational complexity of the mathematical model increases significantly with the size of the analyzed road network. In order to program the problem of the nonlinear traffic flow distribution in this case, it is necessary to use advanced computer software to search for optimal solutions according to the criterion of equal mean times for the Nash equilibrium and equal marginal times for the Stackelberg approach.

7. Conclusions

In summary, the proposed TSMA algorithm was used to develop a transport network model. The presented case study covered three main origin nodes and one destination node. Based on the characteristics of the transport links, a numerical analysis of the traffic flow distribution problem was presented using nonlinear programming. The developed deterministic model of the transport system made it possible to optimize the distribution of the traffic flows within the road network. Minimizing travel time along assumed movement routes was adopted as the objective function for both approaches: the Nash equilibrium and the Stackelberg equation. The analysis presented in this paper has highlighted the utility of the proposed TSMA algorithm and nonlinear programming based on Nash equilibrium and the Stackelberg approach in solving optimisation problems related to the road transport networks.

Based on three scenarios, it was found that the proposed approaches make it possible to accurately determine the optimal distribution of traffic flows in the transport network according to the adopted criteria. Increased traffic volume on individual routes in the transport system has been observed to lead to a decrease in traffic volume on some links of the road network. In the analyzed road network, the increase in traffic on the link (6.7) causes congestion and significant travel delays. This means changes in the distribution of traffic flows for the O-D pair (2,8) between the route containing the link (6,7) and the routes composed of other connections. Therefore, the p_5 route is taken by a small number of traffic users, which results in a reduction in traffic on the (5,6) link, which is only part of the p_5 route. This phenomenon is not intuitive and requires examining the elements that make up road networks in a systems approach, taking into account the complex mechanisms of traffic flow.

A drawback of the developed transport system model is its deterministic character and the failure to take into account planned and random events that affect the distinguishing functions of the elements of the road network. Such important events include collisions, road accidents, road infrastructure renovations, and extremely adverse weather conditions. Therefore, in the next stage of the research, we will try to develop a dynamic model corresponding to a real (variable) road situation. Moreover, we hope to apply fuzzy logic to the decision-making problem, which reflects uncertainty and random components. Expanding the developed transport system model onto a larger area (than the one discussed in the paper) of the Warsaw agglomeration might also be also an appropriate direction of research.

Author Contributions: Conceptualization, M.O.; methodology, M.O. and J.Z.; software, M.O.; validation, M.O. and A.L.; formal analysis, M.O.; investigation, M.O.; resources, M.O.; data curation, M.O. and J.Z.; writing—original draft preparation, M.O.; writing—review and editing, M.O., J.Z., J.M. and A.L.; visualization, M.O.; supervision, J.Z. and J.M.; project administration, M.O. and J.M.; funding acquisition, J.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: This research was funded by the Military University of Technology, grant number UGB 22-835. This support is gratefully acknowledged.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Table A1. Abbreviations and acronyms.

Abbreviation/Acronym	Meaning
EC-TEP	Environmentally constrained traffic equilibrium problem
GDDKiA	General Directorate for National Roads and Motorways
GPR	General Traffic Measurement
MST	Transport system model
O-D	Origin–Destination
PoA	Price of Anarchy
TSMA	Transport system model creation algorithm

Table A2. Notations.

Notation	Definition	
A	Set of origin nodes	
В	Set of destination nodes	
D^{dop}	Set of permissible solutions	
Ε	Set of O-D pairs	
F	Set of structure graph node and (or) link functions	
F_L	Set of functions defined on a set of links L	
F_W	Set of functions defined on a set of nodes W	
G	System structure graph	
L	Set of graph links G	
0	Transport system traffic organization	
P	Traffic flow in MST	
P^{ab}	Set of routes for (a, b) O-D pair	
S	System structure graphs	
T	Time characteristic matrix	
T_c	Total traffic flow movement time	
V	Set of intermediate nodes	
W	Set of graph nodes G	
W	Population of the set W	
X	Traffic flow distribution matrix	
X^*	Traffic flow optimal distribution matrix	
α^{ab}	Minimum travel time within a O-D pair (a, b)	
$a_{p,ab}^{ij}$	1 if route $p \in \underline{P^{ab}}$ contains link $(i, j) \in L$, 0 otherwise	
$\delta^{p,ab}$	Flow rate of O-D pair (<i>a</i> , <i>b</i>) \in <i>E</i> routed on path $p \in \underline{P^{ab}}$	
l _{ii}	Link between w_i and w_i nodes	
$m_{ij}(x_{ij})$	Marginal travel time by link (i, j)	
$m^{p,ab}$	Marginal travel time by route p within a O-D pair (a, b)	
р	Route	
$t^{p,ab}$	Average travel time by route p within a O-D pair (a , b)	
Xii	Traffic flow volume on link (i, i)	
ij	Traffic flow volume on read <i>n</i> within a Q-D pair (a, b)	
A' '	Traine now volume on road p within a O-D pair (u, v)	

Appendix B

Code of the Nash equilibrium (scenario #1) for LINGO 18.0 t14=5+0.05*x14+0.025*(x14²); t25=10+0.04*x25+0.065*(x25²); t36=6+0.03*x36+0.025*(x36²); t47=4+0.04*x47+0.04*(x47²);

t48=5+0.08*x48+0.125*(x48^2); t54=4+0.035*x54+0.02*(x54²); t56=5+0.035*x56+0.04*(x56²); t67=1+0.035*x67+0.03*(x67²); t78=2+0.025*x78+0.035*(x78²); t1=t14+t48; t2=t14+t47+t78; t3=t25+t54+t48; t4=t25+t54+t47+t78; t5=t25+t56+t67+t78; t6=t36+t67+t78; x14>=0;x25>=0; x36>=0; x47>=0; x48>=0; x54>=0; x56 >= 0;x67>=0; x78>=0; x1=4; x2=5; x3=7; x18=4; x28=5; x38=7; x18=x118+x218; x28=x328+x428+x528; x38=x638; x1=x14; x2=x25; x3=x36; x14+x54=x47+x48;x25=x54+x56; x36+x56=x67; x47+x67=x78;x48+x78=x14+x25+x36; x8=x48+x78; x14=x118+x218; x25=x328+x428+x528; x36=x638; x47=x218+x428; x48=x118+x328; x54=x328+x428; x56=x528; x67=x528+x638;x78=x218+x428+x528+x638; t1=t2; t3=t4; t4=t5;

26 of 30

Appendix C

Code of the Stackelberg approach (scenario #1) for LINGO 18.0

```
m14=5+2*0.05*x14+3*0.025*x14^2;
m25=10+2*0.04*x25+3*0.065*x25<sup>2</sup>;
m36=6+2*0.03*x36+3*0.025*x36<sup>2</sup>;
m47=4+2*0.04*x47+3*0.04*x47^2;
m48=5+2*0.08*x48+3*0.125*x48^2;
m54=4+2*0.035*x54+3*0.02*x54<sup>2</sup>;
m56=5+2*0.035*x56+3*0.04*x56^2;
m67=1+2*0.035*x67+3*0.03*x67^2;
m78=2+2*0.025*x78+3*0.035*x78<sup>2</sup>;
m1=m14+m48;
m2=m14+m47+m78;
m3=m25+m54+m48;
m4=m25+m54+m47+m78;
m5=m25+m56+m67+m78;
m6=m36+m67+m78;
x14>=0;
x25>=0;
x36>=0;
x47>=0;
x48>=0;
x54>=0;
x56>=0;
x67>=0;
x78>=0;
x1=4;
x2=5;
x3=7;
x18=4;
x28=5;
x38=7;
x18=x118+x218;
x28=x328+x428+x528;
x38=x638;
x1=x14;
x2=x25;
x3=x36;
x14+x54=x47+x48;
x25=x54+x56;
x36+x56=x67;
x47+x67=x78;
x48+x78=x14+x25+x36;
x8=x48+x78;
x14=x118+x218;
x14=x118+x218;
x25=x328+x428+x528;
x36=x638;
x47=x218+x428;
x48=x118+x328;
x54=x328+x428;
```

```
x56=x528;
x67=x528+x638;
x78=x218+x428+x528+x638;
m1=m2;
m3=m4;
m4=m5;
```

Appendix D



Figure A1. Nguyen–Dupuis network [67].

References

- 1. Macioszek, E.; Staniek, M.; Sierpiński, G. Analysis of Trends in Development of Freight Transport Logistics Using the Example of Silesian Province (Poland)—A Case Study. *Transp. Res. Procedia* 2017, 27, 388–395. [CrossRef]
- Naumov, V.; Szarata, A.; Vasiutina, H. Simulating a Macrosystem of Cargo Deliveries by Road Transport Based on Big Data Volumes: A Case Study of Poland. *Energies* 2022, 15, 5111. [CrossRef]
- 3. Zwolińska, B.; Michlowicz, E. Impact of Change in the Structure of Distribution System on Incurred Cost. *Arch. Transp.* 2016, *39*, 87–94. [CrossRef]
- Zieja, M.; Ziółkowski, J.; Oszczypała, M. Comparative Analysis of Available Options for Satisfying Transport Needs Including Costs. In Proceedings of the 23rd International Scientific Conference Part 3, Palanga, Lithuania, 2–4 October 2019; pp. 1433–1438.
- Ziółkowski, J.; Zieja, M.; Oszczypała, M. Forecasting of the Traffic Flow Distribution in the Transport Network. In Proceedings of the Proceedings of 23rd International Scientific Conference Part 3, Palanga, Lithuania, 2–4 October 2019; pp. 1476–1480.
- Małachowski, J.; Ziółkowski, J.; Oszczypała, M.; Szkutnik-Rogoż, J.; Lęgas, A. Assessment of Options to Meet Transport Needs Using the Maja Multi-Criteria Method. Arch. Transp. 2021, 57, 25–41. [CrossRef]
- Betkier, I.; Żak, J.; Mitkow, S. Parking Lots Assignment Algorithm for Vehicles Requiring Specific Parking Conditions in Vehicle Routing Problem. *IEEE Access* 2021, 9, 161469–161487. [CrossRef]
- 8. Hu, X.; Hao, X.; Wang, H.; Su, Z.; Zhang, F. Research on On-Street Temporary Parking Effects Based on Cellular Automaton Model under the Framework of Kerner's Three-Phase Traffic Theory. *Phys. Stat. Mech. Its Appl.* **2020**, *5*45, 123725. [CrossRef]
- 9. Afrin, T.; Yodo, N. A Survey of Road Traffic Congestion Measures towards a Sustainable and Resilient Transportation System. *Sustainability* **2020**, *12*, 4660. [CrossRef]
- 10. Hu, X.; Qiao, L.; Hao, X.; Lin, C.; Liu, T. Research on the Impact of Entry Points on Urban Arterial Roads in the Framework of Kerner's Three-Phase Traffic Theory. *Phys. Stat. Mech. Appl.* **2022**, *605*, 127962. [CrossRef]
- 11. Angelelli, E.; Morandi, V.; Savelsbergh, M.; Speranza, M.G. System Optimal Routing of Traffic Flows with User Constraints Using Linear Programming. *Eur. J. Oper. Res.* 2021, 293, 863–879. [CrossRef]
- 12. Raza, A.; Ali, M.U.; Ullah, U.; Fayaz, M.; Alvi, M.J.; Kallu, K.D.; Zafar, A.; Nengroo, S.H. Evaluation of a Sustainable Urban Transportation System in Terms of Traffic Congestion—A Case Study in Taxila, Pakistan. *Sustainability* **2022**, *14*, 12325. [CrossRef]
- 13. Li, X.; Li, L.; Huang, B.; Dou, H.; Yang, X.; Zhou, T. Meta-Extreme Learning Machine for Short-Term Traffic Flow Forecasting. *Appl. Sci.* **2022**, *12*, 12670. [CrossRef]
- 14. Allaire, G. Numerical Analysis and Optimization: An Introduction to Mathematical Modelling and Numerical Simulation; OUP: Oxford, UK, 2007; ISBN 978-0-19-152552-0.

- 15. Fowler, A.C.; Fowler, A.C. *Mathematical Models in the Applied Sciences;* Cambridge University Press: Cambridge, UK, 1997; ISBN 978-0-521-46703-2.
- Wasiak, M.; Jacyna-Gołda, I.; Markowska, K.; Jachimowski, R.; Kłodawski, M.; Izdebski, M. The Use of a Supply Chain Configuration Model to Assess the Reliability of Logistics Processes. *Ekspolatacja Niezawodn. Maint. Reliab.* 2019, 21, 367–374. [CrossRef]
- Paś, J.; Klimczak, T.; Rosiński, A.; Stawowy, M. The Analysis of the Operational Process of a Complex Fire Alarm System Used in Transport Facilities. *Build. Simul.* 2022, 15, 615–629. [CrossRef]
- 18. Jang, S.; Wu, S.; Kim, D.; Song, K.-H.; Lee, S.M.; Suh, W. Impact of Lowering Speed Limit on Urban Transportation Network. *Appl. Sci.* 2022, 12, 5296. [CrossRef]
- 19. Izdebski, M.; Jacyna-Gołda, I.; Gołda, P. Minimisation of the Probability of Serious Road Accidents in the Transport of Dangerous Goods. *Reliab. Eng. Syst. Saf.* **2022**, *217*, 108093. [CrossRef]
- Olayode, I.O.; Tartibu, L.K.; Alex, F.J. Comparative Study Analysis of ANFIS and ANFIS-GA Models on Flow of Vehicles at Road Intersections. *Appl. Sci.* 2023, 13, 744. [CrossRef]
- Khudov, H.; Koval, V.; Khizhnyak, I.; Bridnia, Y.; Chepurnyi, V.; Prykhodko, S.; Oleksenko, O.; Glukhov, S.; Savran, V. The Method of Transport Logistics Problem Solving by the MAX-MIN ACO Algorithm. *Int. J. Emerg. Technol. Adv. Eng.* 2022, 12, 108–119. [CrossRef]
- Chen, D.; Cheng, S.; Liu, J.; Zhang, J.; You, X. A Simulation-Based Optimization Method for Truck-Prohibit Ramp Placement along Freeways. *Math. Probl. Eng.* 2023, 2023, 4170669. [CrossRef]
- Yu, R.; Yun, L.; Chen, C.; Tang, Y.; Fan, H.; Qin, Y. Vehicle Routing Optimization for Vaccine Distribution Considering Reducing Energy Consumption. *Sustainability* 2023, 15, 1252. [CrossRef]
- 24. Caban, J.; Droździel, P. Traffic Congestion in Chosen Cities of Poland. *Sci. J. Silesian Univ. Technol. Ser. Transp.* 2020, 108, 5–14. [CrossRef]
- 25. Zhang, W.; Wang, X.; Yang, K. Incentive Contract Design for the Water-Rail-Road Intermodal Transportation with Travel Time Uncertainty: A Stackelberg Game Approach. *Entropy* **2019**, *21*, 161. [CrossRef]
- 26. Guo, J.; Xie, Z.; Li, Q. Stackelberg Game Model of Railway Freight Pricing Based on Option Theory. *Discrete Dyn. Nat. Soc.* **2020**, 2020, 6436729. [CrossRef]
- 27. Pisarski, D.; Canudas-de-Wit, C. Nash Game-Based Distributed Control Design for Balancing Traffic Density Over Freeway Networks. *IEEE Trans. Control Netw. Syst.* 2016, *3*, 149–161. [CrossRef]
- Zhang, X.; Zhang, H.M.; Huang, H.-J.; Sun, L.; Tang, T.-Q. Competitive, Cooperative and Stackelberg Congestion Pricing for Multiple Regions in Transportation Networks. *Transportmetrica* 2011, 7, 297–320. [CrossRef]
- Krichene, W.; Reilly, J.D.; Amin, S.; Bayen, A.M. Stackelberg Routing on Parallel Networks with Horizontal Queues. *IEEE Trans. Autom. Control* 2014, 59, 714–727. [CrossRef]
- Belov, A.; Mattas, K.; Makridis, M.; Menendez, M.; Ciuffo, B. A Microsimulation Based Analysis of the Price of Anarchy in Traffic Routing: The Enhanced Braess Network Case. J. Intell. Transp. Syst. 2022, 26, 448–460. [CrossRef]
- 31. Andelman, N.; Feldman, M.; Mansour, Y. Strong Price of Anarchy. *Games Econ. Behav.* **2009**, *65*, 289–317. [CrossRef]
- 32. Perakis, G. The "Price of Anarchy" Under Nonlinear and Asymmetric Costs. *Math. Oper. Res.* 2007, 32, 614–628. [CrossRef]
- Christodoulou, G.; Koutsoupias, E. The Price of Anarchy of Finite Congestion Games. In Proceedings of the Thirty-Seventh Annual ACM Symposium on Theory of Computing—STOC '05, Baltimore, MD, USA, 22–24 May 2005; ACM Press: Baltimore, MD, USA, 2005; pp. 67–73.
- Lazar, D.A.; Coogan, S.; Pedarsani, R. The Price of Anarchy for Transportation Networks with Mixed Autonomy. In Proceedings of the 2018 Annual American Control Conference (ACC), Milwaukee, WI, USA, 27–29 June 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 6359–6365.
- 35. Youn, H.; Gastner, M.T.; Jeong, H. Price of Anarchy in Transportation Networks: Efficiency and Optimality Control. *Phys. Rev. Lett.* **2008**, *101*, 128701. [CrossRef]
- 36. Zhang, J.; Pourazarm, S.; Cassandras, C.G.; Paschalidis, I.C. The Price of Anarchy in Transportation Networks: Data-Driven Evaluation and Reduction Strategies. *Proc. IEEE* 2018, *106*, 538–553. [CrossRef]
- Harris, I.; Naim, M.; Palmer, A.; Potter, A.; Mumford, C. Assessing the Impact of Cost Optimization Based on Infrastructure Modelling on CO2 Emissions. *Int. J. Prod. Econ.* 2011, 131, 313–321. [CrossRef]
- Liu, Z.; Niu, Y.; Guo, C.; Jia, S. A Vehicle Routing Optimization Model for Community Group Buying Considering Carbon Emissions and Total Distribution Costs. *Energies* 2023, 16, 931. [CrossRef]
- Wróblewski, P.; Lewicki, W. A Method of Analyzing the Residual Values of Low-Emission Vehicles Based on a Selected Expert Method Taking into Account Stochastic Operational Parameters. *Energies* 2021, 14, 6859. [CrossRef]
- Ziółkowski, J.; Lęgas, A.; Szymczyk, E.; Małachowski, J.; Oszczypała, M.; Szkutnik-Rogoż, J. Optimization of the Delivery Time within the Distribution Network, Taking into Account Fuel Consumption and the Level of Carbon Dioxide Emissions into the Atmosphere. *Energies* 2022, 15, 5198. [CrossRef]
- Pan, S.; Ballot, E.; Fontane, F. The Reduction of Greenhouse Gas Emissions from Freight Transport by Pooling Supply Chains. Int. J. Prod. Econ. 2013, 143, 86–94. [CrossRef]
- 42. Djenouri, Y.; Belhadi, A.; Srivastava, G.; Lin, J.C.-W. Hybrid Graph Convolution Neural Network and Branch-and-Bound Optimization for Traffic Flow Forecasting. *Future Gener. Comput. Syst.* **2023**, *139*, 100–108. [CrossRef]

- 43. Zeng, H.; Jiang, C.; Lan, Y.; Huang, X.; Wang, J.; Yuan, X. Long Short-Term Fusion Spatial-Temporal Graph Convolutional Networks for Traffic Flow Forecasting. *Electronics* **2023**, *12*, 238. [CrossRef]
- 44. Anjaneyulu, M.; Kubendiran, M. Short-Term Traffic Congestion Prediction Using Hybrid Deep Learning Technique. *Sustainability* **2023**, *15*, 74. [CrossRef]
- Zhao, X.; Hu, L.; Wang, X.; Wu, J. Study on Identification and Prevention of Traffic Congestion Zones Considering Resilience-Vulnerability of Urban Transportation Systems. *Sustainability* 2022, 14, 16907. [CrossRef]
- 46. Singh, V.; Sahana, S.K.; Bhattacharjee, V. Nature-Inspired Cloud–Crowd Computing for Intelligent Transportation System. Sustainability 2022, 14, 16322. [CrossRef]
- Tay, T.; Osorio, C. Bayesian Optimization Techniques for High-Dimensional Simulation-Based Transportation Problems. *Transp. Res. Part B Methodol.* 2022, 164, 210–243. [CrossRef]
- Strnad, I.; Marsetič, R. Differential Evolution Based Numerical Variable Speed Limit Control Method with a Non-Equilibrium Traffic Model. *Mathematics* 2023, 11, 265. [CrossRef]
- Guo, J.; Harmati, I. Evaluating Semi-Cooperative Nash/Stackelberg Q-Learning for Traffic Routes Plan in a Single Intersection. Control Eng. Pract. 2020, 102, 104525. [CrossRef]
- 50. Stein, O.; Sudermann-Merx, N. The Noncooperative Transportation Problem and Linear Generalized Nash Games. *Eur. J. Oper. Res.* 2018, 266, 543–553. [CrossRef]
- Dixit, V.V.; Denant-Boemont, L. Is Equilibrium in Transport Pure Nash, Mixed or Stochastic? *Transp. Res. Part C Emerg. Technol.* 2014, 48, 301–310. [CrossRef]
- 52. Blanchet, A.; Carlier, G. Optimal Transport and Cournot-Nash Equilibria. Math. Oper. Res. 2016, 41, 125–145. [CrossRef]
- 53. Ghosh, B.; Basu, B.; O'Mahony, M. Multivariate Short-Term Traffic Flow Forecasting Using Time-Series Analysis. *IEEE Trans. Intell. Transp. Syst.* **2009**, *10*, 246–254. [CrossRef]
- Zheng, J.; Huang, M. Traffic Flow Forecast Through Time Series Analysis Based on Deep Learning. IEEE Access 2020, 8, 82562–82570. [CrossRef]
- Ding, A.; Zhao, X.; Jiao, L. Traffic Flow Time Series Prediction Based on Statistics Learning Theory. In Proceedings of the IEEE 5th International Conference on Intelligent Transportation Systems, Singapore, 6 September 2002; IEEE: Piscataway, NJ, USA, 2002; pp. 727–730.
- 56. Yu, Q.; Lei, L.; Wang, L. Research on Safety and Traffic Efficiency of Mixed Traffic Flows in the Converging Section of a Super-Freeway Ramp. *Sustainability* **2022**, *14*, 13234. [CrossRef]
- 57. Cascetta, E.; Inaudi, D.; Marquis, G. Dynamic Estimators of Origin-Destination Matrices Using Traffic Counts. *Transp. Sci.* **1993**, 27, 363–373. [CrossRef]
- Wu, J.; Chang, G.-L. Estimation of Time-Varying Origin-Destination Distributions with Dynamic Screenline Flows. *Transp. Res.* Part B Methodol. 1996, 30, 277–290. [CrossRef]
- 59. Wei, W.; Wu, Q.; Wu, J.; Du, B.; Shen, J.; Li, T. Multi-Agent Deep Reinforcement Learning for Traffic Signal Control with Nash Equilibrium. In Proceedings of the 2021 IEEE 23rd Int. Conf. on High Performance Computing & Communications; 7th Int Conf on Data Science & Systems; 19th Int Conf on Smart City; 7th Int Conf on Dependability in Sensor, Cloud & Big Data Systems & Application (HPCC/DSS/SmartCity/DependSys), Haikou, China, 20–22 December 2021; pp. 1435–1442.
- 60. Kyriakou, K.; Lakakis, K.; Savvaidis, P.; Basbas, S. Analysis of Spatiotemporal Data to Predict Traffic Conditions Aiming at a Smart Navigation System for Sustainable Urban Mobility. *Arch. Transp.* **2019**, *52*, 27–46. [CrossRef]
- Fabrikant, A.; Papadimitriou, C.; Talwar, K. The Complexity of Pure Nash Equilibria. In Proceedings of the Thirty-Sixth Annual ACM Symposium on Theory of Computing—STOC '04, Chicago, IL, USA, 13–15 June 2004; ACM Press: New York, NY, USA, 2004; p. 604.
- 62. Yu, H.; Fang, J.; Liu, S.; Ren, Y.; Lu, J. A Node Optimization Model Based on the Spatiotemporal Characteristics of the Road Network for Urban Traffic Mobile Crowd Sensing. *Veh. Commun.* **2021**, *31*, 100383. [CrossRef]
- 63. Holt, C.A.; Roth, A.E. The Nash Equilibrium: A Perspective. Proc. Natl. Acad. Sci. USA 2004, 101, 3999–4002. [CrossRef]
- 64. Abdelghaffar, H.M.; Yang, H.; Rakha, H.A. Isolated Traffic Signal Control Using Nash Bargaining Optimization. *Glob. J. Res. Eng. B Automot. Eng.* **2016**, *16*, 1–11.
- 65. Jahn, O.; Möhring, R.H.; Schulz, A.S.; Stier-Moses, N.E. System-Optimal Routing of Traffic Flows with User Constraints in Networks with Congestion. *Oper. Res.* 2005, *53*, 600–616. [CrossRef]
- 66. Fisk, C.S. Game Theory and Transportation Systems Modelling. Transp. Res. Part B Methodol. 1984, 18, 301–313. [CrossRef]
- 67. Nguyen, S.; Dupuis, C. An Efficient Method for Computing Traffic Equilibria in Networks with Asymmetric Transportation Costs. *Transp. Sci.* **1984**, *18*, 185–202. [CrossRef]
- 68. Xu, X.; Chen, A.; Cheng, L. Reformulating Environmentally Constrained Traffic Equilibrium via a Smooth Gap Function. *Int. J. Sustain. Transp.* **2015**, *9*, 419–430. [CrossRef]
- Delle Site, P. Pricing of Connected and Autonomous Vehicles in Mixed-Traffic Networks. *Transp. Res. Rec. J. Transp. Res. Board* 2021, 2675, 178–192. [CrossRef]
- Yang, X.; Ban, X.J.; Ma, R. Mixed Equilibria with Common Constraints on Transportation Networks. Netw. Spat. Econ. 2017, 17, 547–579. [CrossRef]
- 71. Zhou, B.; Song, Q.; Zhao, Z.; Liu, T. A Reinforcement Learning Scheme for the Equilibrium of the In-Vehicle Route Choice Problem Based on Congestion Game. *Appl. Math. Comput.* **2020**, *371*, 124895. [CrossRef]

72. Lin, H. Responsive Transport Network Design: Minimal Investment for Desired Travel Time Reduction. *Transp. Lett.* **2022**, *14*, 651–659. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.